Author's response to reviews

Title: Macrolide resistance determinants among Streptococcus pneumoniae isolates from carriers in Central Greece

Authors:

Ioanna N Grivea (iogrivea@gmail.com)
Alexia Sourla (alexia.sourla@gmail.com)
Eleni Ntokou (eleni.ntokou@gmail.com)
Denise C Chryssanthopoulou (denisec@otenet.gr)
Alexandra G Tsantouli (atsantouli@hotmail.com)
George A Syrogiannopoulos (syrogian@otenet.gr)

Version: 4 Date: 26 September 2012

Author's response to reviews: see over
September 26, 2012

Editor-in-Chief
BMC Infectious Diseases

Dear Sir,

We are herewith submitting the revised manuscript entitled “Macrolide resistance determinants among Streptococcus pneumoniae isolates from carriers in Central Greece”. We have highlighted the changes made for the current 2nd revision.

We greatly appreciate your meticulous work.

Sincerely yours,

George A. Syrogiannopoulos, MD
Professor of Pediatrics